 CHOHAN /10--Oct- 05

Week 41 starting 10 oct 05
SM18 Operation Meeting (Tuesdays)

Bench status  on Monday  10 Oct 05 : 10/10/2005 8:47:15 PM   
	1
	A1
	MBBL2216
	PT 10 Min. NRJ 11850 A
	1.90

	7
	A2
	MBAL1249
	ICS 4 Final connection
	294.28

	3
	B1
	SSS629
	Prep 5 Cool Down
	18.88

	8
	B2
	SSS349
	Prep 3 CDW1 HV
	295.8

	11
	C1
	MBBR1263
	PT 13.2 Resist. Meas.
	299.2

	4
	C2
	MBBL2071
	Prep 5 Cool Down
	98.13

	2
	D1
	SSS249
	PT 12 Warm Up
	273.06

	10
	D2
	SSS630
	Finished Ready to go Final scan
	295.18

	6
	E1
	MBAR2229
	Prep 5 Cool Down
	267.56

	12
	E2
	CRYO
	Cryo leak repair
	295.62

	9
	F1
	MBBL1189
	Prep 3 CDW1 HV
	294.1

	5
	F2
	MBAR1253
	Prep 5 Cool Down
	245.47


Tempr. from TBSM18 site
   
 
   
Coordinator this week: Marta; MTM support: 16-3602

15 complete Magnet Tests cycles  from 11 benches ( E2 under CFB repair)
 10 dipoles + 3  SSS  + 2  SSSS
 Disconnected or ROBTC after tests last week ( Sat Midnight to Sat midnight) 
Situation From Cryogenics:  J-P Lamboy or ACR Rep.
[ CRYO PROBLEMS SAT & SUN ? Cooldown of sss629 on B1 ?]

E2 CFB (leak) Repair whole Week this week again?  ( 11 benches usage only)
 Special  Points :
· Special SSS on B1 : ( ) 
· Special SSS on D2 :Tests Plans ?

· AOB
15 Magnets this week. Only 11 Benches.( E 2 under CFB leak repair) 2 benches with special SSSS
+ at least 2 magnets more were finished on Saturday(sss160 & sss630)  BUT ICS had STAFF problems ( only 1 person on shift)

Status Sun Morning09/10/2005 11:40:08  
	3
	A1
	MBBL2216
	PT 2 HV Cold Test
	4.39

	8
	A2
	MBBR1261
	ICS 6 Disconnect Magnet WP07
	294.73

	11
	B1
	SSS629
	CR 1 Leak Test
	294.55

	9
	B2
	SSS160
	PT 14 Magnet MAPS
	295.8

	2
	C1
	MBBR1263
	PT 6.6 Training 6
	4.95

	4
	C2
	MBBL2071
	ICS 4 Final connection
	294.64

	5
	D1
	SSS249
	Prep 5 Cool Down
	5.02

	7
	D2
	SSS630
	PT 14 Magnet MAPS
	295.18

	10
	E1
	MBAR2229
	Prep 3 CDW1 HV
	295.8

	12
	E2
	CRYO
	Cryo leak repair
	295.09

	1
	F1
	MBAL3643
	PT 13.2 Resist. Meas.
	296.88

	6
	F2
	MBAR1253
	ICS 4 Final connection
	293.12

	

	

	  Magnets under Test from 02-10-2005 to 08-10-2005 

	-
	Bench
	Magnet
	Shafts
	+
	Start Date
	Connect
Time
	Cooling
Time
	Cold Test
Time
	Warming
Time
	Disconnect
Time
	End Date
	Total
	Signed for
	
	

	1
	TBA1
	MBBR3389 
	[image: image1.png]



	


	01-10-2005 
	49 h.
	23 h.
	47 h.
	14 h.
	12 h.
	07-10-2005 
	145 h.
	Stripping
	
	

	2
	TBA2
	MBAL3400 
	[image: image3.png]



	


	29-09-2005 
	26 h.
	21 h.
	20 h.
	16 h.
	11 h.
	03-10-2005 
	94 h.
	Stripping
	
	

	3
	TBB1
	SSS642 
	[image: image5.png]



	


	20-09-2005 
	124 h.
	14 h.
	144 h.
	24 h.
	25 h.
	05-10-2005 
	331 h.
	Stand By
	
	

	4
	TBB2
	SSS167 
	[image: image7.png]



	


	01-10-2005 
	34 h.
	18 h.
	17 h.
	9 h.
	8 h.
	05-10-2005 
	86 h.
	Stripping
	
	

	5
	TBC1
	MBAL3734 
	[image: image9.png]



	


	02-10-2005 
	31 h.
	31 h.
	18 h.
	15 h.
	15 h.
	06-10-2005 
	110 h.
	Stripping
	
	

	6
	TBC2
	MBAL2064 
	[image: image11.png]



	


	02-10-2005 
	61 h.
	20 h.
	15 h.
	15 h.
	7 h.
	07-10-2005 
	118 h.
	Stand By
	
	

	7
	TBC2
	MBBL1214 
	[image: image13.png]



	


	30-09-2005 
	39 h.
	3 h.
	0 h.
	8 h.
	0 h.
	02-10-2005 
	50 h.
	Stand By
	
	

	8
	TBD1
	SSS091 
	[image: image15.png]



	


	30-09-2005 
	35 h.
	12 h.
	37 h.
	9 h.
	29 h.
	05-10-2005 
	122 h.
	Stripping
	
	

	9
	TBE1
	MBBL2203 
	[image: image17.png]



	


	01-10-2005 
	34 h.
	27 h.
	4 h.
	17 h.
	11 h.
	04-10-2005 
	93 h.
	Stripping
	
	

	10
	TBE1
	MBBL3403 
	[image: image19.png]



	


	04-10-2005 
	29 h.
	23 h.
	16 h.
	17 h.
	9 h.
	08-10-2005 
	94 h.
	Stripping
	
	

	11
	TBF1
	MBBL2204 
	[image: image21.png]



	


	01-10-2005 
	33 h.
	23 h.
	15 h.
	18 h.
	17 h.
	05-10-2005 
	106 h.
	Stripping
	
	

	12
	TBF2
	MBAL3396 
	[image: image23.png]



	


	04-10-2005 
	24 h.
	24 h.
	18 h.
	16 h.
	9 h.
	08-10-2005 
	91 h.
	Stripping
	
	

	13
	TBF2
	MBBR1257 
	[image: image25.png]



	


	30-09-2005 
	32 h.
	25 h.
	14 h.
	16 h.
	18 h.
	04-10-2005 
	105 h.
	Stripping
	
	


Quench Information
	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBBR3389
	9066 A.
(D2_U)
	10221 A.
(D1_U)
	11978 A.
(D1_U)
	12350.4 A.
(D1-L)
	12734 A.
(D2-U)
	12850 A.
[image: image27.jpg]



	-
	-
	-
	-
	-
	-
	-

	2
	MBAL3400
	11098.5 A.
(D2-L)
	11928.9 A.
(D1-U)
	12680.6 A.
(D2-U)
	12850 A.
[image: image28.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-

	3
	SSS642
	5799 A.
(MQMC_Int)
	5800 A.
[image: image29.jpg]



	4650 A.
[image: image30.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-

	4
	SSS167
	12179.3 A.
(QE12)
	12429.3 A.
(QE34)
	12541.8 A.
(QE_34)
	[image: image31.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	

	5
	MBAL3734
	12119.3 A.
(D2_L)
	12818.5 A.
(D2_U)
	12850 A.
[image: image32.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	

	6
	MBAL2064
	11795.5 A.
(D1-L)
	12157 A.
(D1_U)
	12850 A.
[image: image33.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	

	7
	MBBL1214
	[image: image34.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	

	8
	SSS091
	11319 A.
(QI_34)
	11459 A.
(QI_34)
	11850 A.
(QE_12)
	11923.3 A.
(QI_12)
	11986.8 A.
(QI_12)
	12084.1 A.
(QE-34)
	12551.2 A.
(QE-12)
	12747 A.
(QE-12)
	12850 A.
[image: image35.jpg]



	-
	-
	-
	-
	

	9
	MBBL2203
	[image: image36.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	

	10
	MBBL3403
	12569.3 A.
(D2-L)
	12850 A.
[image: image37.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	

	11
	MBBL2204
	11973 A.
(D1-L)
	12472 A.
(D1-L)
	12818 A.
(D1_L)
	[image: image38.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	

	12
	MBAL3396
	10003.1 A.
(D2-U)
	12325.1 A.
(D2-L)
	11996 A.
(D2-U)
	12159 A.
(D2-L)
	12303 A.
(D1-U)
	[image: image39.png]



	-
	-
	-
	-
	-
	-
	-
	

	13
	MBBR1257
	10613.7 A.
(D1_U)
	12171 A.
(D1_U)
	12493.8 A.
(D1-U)
	[image: image40.png]



	-
	-
	-
	-
	-
	-
	-
	-
	-
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Shows  SSSS work on B1  including painful 10 wire breaks in SSW ; But also long period in connection before cooldown
	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday_27_A
	TBF2


	MB3396: wrong TT821 value.

	CRI
	
	NC opened. HWREC put off. Missing data in MTF. It will be the case for the next Noell magnet

	Tuesday_04_N
	TBC1


	MB3734 , first training quench. The hfoff  analysis show an inverted signal for D1-D2

	MTM
	HWREC
	Maryline called. Configuration with HWREC was not properly done.

	Tuesday_04_N
	TBD2


	SSL test at 4.55 K for SSS630: analysis of first quench shows a peak voltage out of limit.

	MTM
	
	Salem contacted. He advised to continue.

	Tuesday_04_N
	TBE1


	MB3403: wrong TT821 value.

	MTM
	
	Test (IAP@warm) stopped till HWREC is put off.

	Tuesday_04_N
	TBA1


	MB3389: Shaft in aperture 1 not possible to insert.

	ICS
	
	Stephane asked to 

· proceed with the MM only in aperture 2

· to Juan and Jurek to investigate the origin of the problem

From Leica measurements it was found that the MRB is laterally displaced by 5 mm.

	Wednesday_05_A
	TBB1


	SSS631 not conform: one corrector dipole circuit was found opened at warm.

	CRI
	
	Connection stopped. Magnet removed from the bench and sent back to 904.

	Thursday_06_M
	TBD2


	SSW measurements values not correct.
	MTM
	
	Guy found that the current reading in not correct (10 times lower). Config. Corrected in the Data Base then OK.

	Thursday_06_N
	TBC1


	Megger in MY93 rack doen't work properly. It delivers around 700V instead of 600V asked. 
	MTM
	
	To be repaired. Action Richard. 

	Thursday_06_N
	TBC1


	Bad communication between control room sun-station and local sun station during HV test and IAP@warm . 
	AB/CO
	
	Logout on the both sides and re-launch TEMA

	Thursday_06_N
	TBE


	IAP @ cold on MB3403: the following potaim cards could not be adjusted:

-Vgas_int_lead

-D1_L-U.


	MTM
	
	Potentiometer has no effect. Potaim card Vgas_int_lead changed by Maryline.

	Friday_07_M
	TBC2


	Resistance value of the cryo_heater in MB2064 indicated that the circuit is opened. 
	MTM
	
	NC opened?

	Friday_07_M
	TBB2


	LF could not be armed while launching the powering of the correctors of the SSS160.
	ACR/MTM
	
	Maryline informed. E_clamp was found broken in the CFB durin the cool-down. A short has been done on the rear of the POTAIM crate.

	Friday_07_N
	TBD1


	LF DAQ could not be triggered automatically during prep2.2 C (test of correctors)..
	MTM
	
	Maryline advised to trigger the DAQ manually.

	Sunday_09_M
	ALL


	Not possible to cool down magnet below 80 K. Communication problem in the process,
	ACR
	
	Cold test stopped for 4 hours.

	Sunday_09_N
	TBD1


	Quenches (5 times) on the corrector CD circuit. No file because card disabled.
	MTM
	
	Maryline contacted. She enables the card.

	Sunday_09_N
	TBB1


	Alarm on anticrysotat of SSS629.
	MTM
	
	Maryline contacted. She asked to load the corresponding parameters for the new AC.

	Sunday_09_N
	TBB2


	Trigger on V_ E_clamp during IAP@warm for SSS349
	ICS?
	
	Giorgio checked with a bridge and electronics is o.k. ICS was asked to check the connections.

	Sunday_09_N
	TBA1


	LHC cycle could not be lauched. Missing files in HWREC.
	MTM
	
	The wrong cycle was selected in the extended test program. Correct cycle was chosen and then it worked.

	Sunday_09_N
	TBA1


	LHC cycle on MB2216 stopped because of a cryo loss.
	ACR
	
	Flow of current lead went down below the threshold. Protection were activated..











